Author's response to reviews

Title: A social marketing approach to implementing evidence-based practice in the VHA QUERI: The TIDES depression collaborative care model

Authors:

Jeff Luck MBA, PhD (jluck@ucla.edu)
F A Hagigi DrPH, MBA, MPH (hagigi@ph.ucla.edu)
Louise E Parker PhD (parkerlouise@earthlink.net)
Elizabeth M Yano PhD, MSPH (elizabeth.yano@va.gov)
Lisa V Rubenstein MD, MSPH (lisar@rand.org)
JoAnn E Kirchner MD (kirchnerjoanne@uams.edu)

Version: 5 Date: 10 August 2009

Author's response to reviews:

The vast majority of suggested changes were acceptable. I responded to the copy editor's comment and made some minor changes to preserve meaning:

p 1: Updated FH's e-mail address

p 2, Abstract, Results, 1st Line: Revised edit to preserve meaning.

pp 2, 8, 9, 14, 20, 22, 31, 34, 36, 37: Spelled "veterans" without initial capital where it is not used in a title.

p 8, 2nd para, last sentence: Revised edit to preserve meaning.

p 10, 3rd-to-last line: Changed "Check" to "Study" and defined "PDSA."

p 12, Comment: The editor is correct - should be PDSA.

p 12, last para, 2nd Line: Revised edit to preserve meaning.